Author’s response to reviews

Title: Actions of sesquiterpene lactones isolated from Moquiniastrum polymorphum subsp. floccosum in MCF7 cell line and their potentiating action on doxorubicin

Authors:

Mariana Mauro (mari_mauro84@hotmail.com)
Renata Matuo (matuorenata@yahoo.com.br)
Natan de David (natan.dedavid@gmail.com)
Regiane Strapasson (regi_0112@hotmail.com)
Rodrigo Juliano Oliveira (rodrigo.oliveira@ufms.br)
Maria Élida Stefanello (elida@ufpr.br)
Cândida Aparecida Kassuya (candida2005@gmail.com)
Maria de Fátima Matos (matosmfc@gmail.com)
Fábio José Faria (fabio.faria@ufms.br)
Deiler Costa (deilercosta@yahoo.com.br)

Version: 2 Date: 02 Jun 2017

Author’s response to reviews:

All editorial revisions requested by the editor were answered as described below:

Request 1. Please include a title page to your manuscript is properly formatted according to our submission guidelines.

Modification: The title page was added containing the title of the article, e-mail and institutional address of all authors and an indication of the corresponding author.

Request 2. We notice a few typographical errors:

a) Methods section of abstract: please change analyzes to analyzes in first sentence.

b) Availability of data and materials: &quot;are not publicly available due" is repeated twice. Please amend.
Modifications: a) the word "analyzes" in the first sentence was amend to "analyses"

b) The repetition of the sentence was removed.

Request 3. Please provide more justification for the contributions of the following authors in your Authors' contributions section: RM, ND, RLBS, MFCM, and DSC. Currently their contributions do not automatically qualify for authorship, the acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship. Please amend your manuscript accordingly.

Modified: The contributions of authors cited were best described and are in accordance with what is considered an important contribution to the author.